Author's response to reviews

Title: Relationship between blood pressure measurements recorded on patients' charts in family physicians' offices and subsequent 24 hour ambulatory blood pressure monitoring

Authors:

Dr Marshall Godwin (godwinm@post.queensu.ca)
Dianne Delva (mdd2@post.queensu.ca)
Rachelle Seguin (seguinr@post.queensu.ca)
Ian Casson (ric@post.queensu.ca)
Susan MacDonald (sem@post.queensu.ca)
Dr Richard Birtwhistle (birtwhis@post.queensu.ca)
Miu Lam (lamm@post.queensu.ca)

Version: 4 Date: 29 Mar 2004

We have made the editorial changes requested. You had said we should make any further changes that the reviewers had suggested. However you did not provide us with links to any further comments from the reviewers so we were not able to do this. Presumably they were ok with the changes we made after the first iteration.